Author’s response to reviews

Title: Survey of the Knowledge, Attitude and Practice of Nigerian Surgery Trainees to HIV-infected persons and AIDS patients

Authors:

Clement A Adebamowo (cadebamo@hsph.harvard.edu)
Emma R Ezeome (ezeome@infoweb.abs.net)
Johnson A Ajuwon (uchmed@skannet.com)
Temidayo O Ogundiran (toogundiran@yahoo.co.uk)

Version: 4 Date: 28 Aug 2002

HIV+ has been replaced with HIV positive
All instances where digits start sentences have been changed to words.
Results have been reported in past tense and the changes indicated in the text by Dr. Maupome have been accepted.
All numbers indicating percentages have been followed by %.
The abstract has been shorted to 239 words.
References have been indicated in square brackets in the text and the list of references have been written according to the style of BioMed Central.